Author’s response to reviews

Title: Evaluation of microstructural changes in spinal cord of patients with degenerative cervical myelopathy by diffusion kurtosis imaging and investigate the correlation with JOA score

Authors:

ZhuoHang Liu (13894707172@163.com)
BingYang Bian (1241928390@qq.com)
Gang Wang (wanggangletyer@126.com)
Cheukying Tian (Cheuk.Tang@mssm.edu)
ZhenShan Lv (lzsspine@jlu.edu.cn)
ZhiQing Shao (390622965@qq.com)
Dan Li (13604430730@126.com)

Version: 3  Date: 26 Apr 2020

Author’s response to reviews:

1. Please remove the participants ages from the figure legends.
   Answer: we remove the participants ages from the figure legends. Thanks.

2. Please amend the consent statement to read: "written informed consent was obtained from each participant"
   Answer: we amend the consent statement to read: "written informed consent was obtained from each participant". Thanks.